Author's response to reviews

Title: Overexpression of eIF-5A2 in mice causes accelerated organismal aging by increasing chromosome instability

Authors:

Muhan Chen (chenm@cshl.edu)
Jian-Dong Huang (Jdhuang@hkucc.hku.hk)
Hongkui Deng (Hongkui_deng@pku.edu.cn)
Suisui Dong (dongsuisui@gmail.com)
Wen Deng (wend@hkucc.hku.hk)
Sze Lan Tsang (Jdhuang@hkucc.hku.hk)
Michael S. Y. Huen (huen.michael@hku.hk)
Leilei Chen (pollyllc@hku.hk)
Tong Zan (Jdhuang@hkucc.hku.hk)
Gui-Xia Zhu (jennyzhu@hkusua.hku.hk)
Xin-Yuan Guan (xyguan@hkucc.hku.hk)

Version: 6 Date: 12 May 2011

Author's response to reviews:

May 12, 2011

Dr. Pulivathi Rao,
BMC cancer Editorial Office,

Dear Dr Rao,

RE: MS: 1149143646400892

Title: Overexpression of eIF-5A2 in mice causes accelerated organismal aging by increasing chromosome instability

Thank you for the acceptance of our manuscript for publication in Plos One. According to your suggestions, we already revised our manuscript. I hope the changes satisfy the requirements of the Editorial Board. I thank you again for reviewing the manuscript and look forward to hearing your favorable reply soon.

Sincerely yours,

Xin-yuan Guan, Ph.D.

Professor
Director, Laboratory of Cancer Genetics,
Department of Clinical Oncology,
The University of Hong Kong,
Room L10-56, 10/F, Laboratory Block
21 Sassoon Road, Pokfulam, Hong Kong
Tel: (852) 28199782; Fax: (852) 28169126
E-mail: xyguan@hkucc.hku.hk